Author’s response to reviews

Title: In vivo analgesic, anti-inflammatory, and sedative activity and a molecular docking study of dinaphthodiospyrol G isolated from Diospyros lotus

Authors:

Dr. Abdur Rauf (mashaljcs@yahoo.com)
Tareq Abu-Izneid (tizneid@gmail.com)
Fahad A. Alhumaydhi (f.alhumaydhi@qu.edu.sa)
Naveed Muhammad (drnaveedrph@gmail.com)
Abdullah S. M. Aljohani (dr.johani@live.com)
Saima Naz (saima_khan201164@yahoo.com)
Saud Bawazeer (sbawazeer@live.com)
Abdul Wadood (awadood@awkum.edu.pk)
Mohammad S. Mubarak (mmubarak@ju.edu.jo)

Version: 1 Date: 09 Apr 2020

Author’s response to reviews:

1. Funding: Please state clearly the role of funder(s) had in your study in the "funding" section of the declarations.
   Reply: Done